Author’s response to reviews

Title: Treatment outcomes and factors affecting treatment outcomes of new patients with tuberculosis in Busan, South Korea: A retrospective study of a citywide registry, 2014-2015

Authors:

Jeongha Mok (mokgamokga@gmail.com)
Daeseong An (niche03@naver.com)
Seoungjin Kim (ksj830318@gmail.com)
Miyoung Lee (dignya@daum.net)
Changhoon Kim (kchprev@gmail.com)
Hyunjin Son (hjson78@gmail.com)

Version: 1 Date: 25 Jun 2018

Author’s response to reviews:

1. Please confirm that the Institutional Review Board of Pusan National University Hospital waived the need for informed participatory consent?

A1) We have revised the “Method section” as editor’s comments.

    (page 5, second paragraphs)

2. Please clarify whether the dataset observed in this study was private or public. If it is publicly available, please provide a link directly to the dataset. If the dataset is private, please provide clarification on whether any (administrative) permission was required, and obtained, in order to review patient records and use the data mentioned in the Methods. Please also state whether the data analysed had been de-identified. Please include this information in the “Ethics approval and consent to participate” section of the Declarations.
A2) We have revised the “Method section” and “Ethics approval and consent to participate” section as editor’s comments.

   (page 5, second paragraph / page 15)